Author's response to reviews

Title: Detection of novel avian-origin influenza A (H7N9) virus in human by multiplex one-step real-time RT-PCR assay

Authors:

Jian Fan (3319fanjian@163.com)
David Cui (davidcui101@163.com)
Siu Ying Lau (sylau926@hkucc.hku.hk)
Guo Liang Xie (ytxiegouliang@163.com)
Xi Chao Guo (guoxichaooqin926667@126.com)
Shi Guo Zheng (zsfzheng@163.com)
Xiao Feng Huang (stevehxf@hku.hk)
Shi Gui Yang (ysgcdc@163.com)
Xian Zhi Yang (yayayang1984@sina.com)
Zhao Xia Huo (huozhaoxia8253@163.com)
Fei Yu (yf01011@live.cn)
Jian Zhong Lou (justljz@gmail.com)
Li Tian (grapetian@163.com)
Xue Fen Li (zylixf@126.com)
Yue Jiao Dong (candy242313@163.com)
Qiao Yun Zhu (akuzhu@163.com)
Yu Chen (chenyu_zy@163.com)

Version: 2 Date: 27 March 2014

Author's response to reviews: see over
Dear Neil,

We would like to submit our revised manuscript entitled “Detection of novel avian-origin influenza A (H7N9) virus in humans by multiplex real-time RT-PCR assay” to the “BMC Infectious Diseases”. We expect that the manuscript can be considered for acceptance.

According to your good suggestions, we had revised our manuscript.

1. All the email addresses of authors had also included in the ‘Title page’ of our revised manuscript.
2. Line and page numbers had been included in the revised manuscript.
3. In addition, figures and legends were included in the main manuscript at the end of the document.

We thank you for your consideration of our manuscript in your journal.

Sincerely,

Yu Chen

Department of Clinical Laboratory, the First Affiliated Hospital, School of Medicine, Zhejiang University, 79 Qingchun Road, Hangzhou, China, 310003,

chenyu_zy@163.com